Author’s response to reviews

Title: Multifactorial intervention for hip and pelvic fracture patients with mild to moderate cognitive impairment: Study protocol of a dual-centre randomised controlled trial (OF-CARE)

Authors:

Anja Dautel (anja.dautel@rbk.de; adautel@web.de)
Tobias Eckert (tobias.eckert@bethanien-heidelberg.de)
Michaela Gross (michaela.gross@rbk.de)
Klaus Hauer (khauer@bethanien-heidelberg.de)
Martina Schäufele (martina.schaeufele@hs-mannheim.de)
André Lacroix (andre.lacroix@bethanien-heidelberg.de)
Ingrid Hendlmeier (i.hendlmeier@hs-mannheim.de)
Bastian Abel (bastian.abel@bethanien-heidelberg.de)
Rebekka Pomiersky (rebekka.pomiersky@bethanien-heidelberg.de)
Julia Gugenhan (julia.gugenhan@rbk.de)
Gisela Büchele (gisela.buechele@uni-ulm.de)
Katrin C. Reber (k.reber@uke.de)
Clemens Becker (clemens.becker@rbk.de)
Klaus Pfeiffer (klaus.pfeiffer@rbk.de)

Version: 4 Date: 08 Apr 2019

Author’s response to reviews:

Dear Stephen Madamba,

We want to thank for your comments. They were thoroughly reviewed and are listed point-by-point below. The required adjustments were made in the manuscript.

Thank you very much for processing and finalizing our manuscript.
Best regards

Anja Dautel on behalf of all co-authors

1) 'We noticed that the corresponding author in the submission system (Anja Dautel) differs from the corresponding author listed on the title page of the manuscript (Klaus Pfeiffer). Please ensure the corresponding author matches on both the submission system and the manuscript.'

The corresponding author is now Klaus Pfeiffer - in submission system as well as in manuscript.

2) 'Please ensure all authors and their individual contributions are included in the Authors’ Contributions section. Currently, author Katrin C. Reber is not included in this section.'

Adjustments were made in the manuscript (Page 32, line 709 to 710): KR contributed to the planned evaluation of economic data.